Abstract

Trade disputes over national product standards are a growing source of tension in the international trading system. The usual pattern is that a country introduces a new product standard for all sales of a good in its local market, which is justified as necessary for consumer or environmental protection. Importers into the local market, however, challenge the standard as a "disguised barrier to trade" or "green protectionism". The paper develops a two country political economy model to explain such disputes. It is shown how the political process can lead to a "political failure" which takes the form of either too many or too few product standards and disagreement between politicians in different countries over the optimal policy. In a second step the model is used to evaluate whether two common proposals to settle or avoid such disputes, mutual recognition of standards and harmonization, can improve the political process.

Product Standards, Trade Disputes and Protectionism

Daniel Sturm

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

© Daniel Sturm, submitted August 2000

ISBN 0 7530 1445 9

Individual copy price: £5

Product Standards, Trade Disputes and **Protectionism**

Daniel Sturm

1.	Introduction		
2.	The Model		5
	2.1 Preference	s and Technology	5
	2.2 The Exter	= =	6
	2.3 Policy Inst	truments	7
	2.4 Welfare an		8
3.	The Political System		10
		' Preferences	10
	3.2 Timing of	Events and Information Structure	13
	_	of Political Equilibrium	14
4.	Analysis of Equilibrium		14
	•	's Optimal Re-Election Rule	15
		the Politicians	16
		quilibrium in the First Period	17
5.	Institutional Reform		20
	5.1 Mutual Re	ecognition of Standards	20
	5.2 Harmonisa	<u> </u>	21
6.	Discussion		23
7.	Conclusion		
8.	Appendix		
Refe	erences		35

Acknowledgements

This paper forms part of my PhD thesis to be submitted at LSE. I would like to thank Tim Besley, Peter Neary and Tony Venables for their help throughout the project. I am also indebted to Gilles Duranton, Jose Madeira, Niko Matouschek, Pawan Patil, Paolo Ramezzana, Steve Redding, S nje Reiche, Till Requate, Alan Winters and Seminar Participants at LSE, EEA, Verein fuer Socialpolitik, ETSC, Tilburg and Heidelberg for many suggestions and comments. All remaining errors are, of course, mine. Financial support from the European Commission through a Marie Curie Fellowship is gratefully acknowledged.

Daniel Sturm is a member of the Centre for Economic Performance, London School of Economics and the Department of Economics, University of Munich.

1 Introduction

One of the most remarkable economic policy developments in the post war period has been the dramatic reduction in tariff levels. This decline in tariff levels has since the early seventies, however, been partially offset by increases in non-tariff barriers to trade. Attempts to reduce such non-tariff barriers are now at the heart of most trade negotiations. The arguably most difficult issue in this area are trade disputes over national product standards. The usual pattern is that a country introduces a new product standard, which applies to all sales of a particular good in the local market, which is justified as necessary for consumer or environmental protection. Importers to the local market, however, challenge the standard as "a disguised barrier to trade" or simply "green protectionism", with the exact meaning of these phrases clearly being hotly disputed.

One prominent example is the "Beef Hormones" dispute between the European Union and the United States. After allegations that the consumption of beef from cattle that have been treated with growth hormones could pose a risk to human health, the EU banned the sale of all hormone-treated beef in the European Union in January 1989. In the United States the hormones in question continue to be allowed and are widely used. Before the ban US exports of beef to the EU consisted mainly of varietal meats, such as kidneys and tongues, while the rest of the meat was sold on the American market. To supply the European market with hormonefree varietal meat American producers would have had to abandon the use of hormones for a large part of their cattle, which would have substantially reduced their profits on the American market and made this possibility uneconomical. As a consequence American exports of beef to the EU dropped by nearly 80 percent in the year after the ban. The United States has repeatedly challenged the ban as not based on the available scientific evidence and therefore constituting a disguised barrier to trade. The EU defends the ban on the ground that it has the right to implement whatever regulation it considers necessary to protect consumers.¹

Many similar cases have kept trade negotiators busy. The United States and

¹For a more detailed summary of this and other trade disputes over product standards see Vogel (1995). Esty (1994) also provides short summaries and Engel (1999) contains an extensive case study of the BSE crisis and a dispute between Venezuela and the United States.

Japan have, for example, for a long time argued about appropriate standards for food additives. Similarly, the member countries of the European Union had an intense struggle over emission standards for cars. The continuous introduction of new production technologies, such as for example genetically-modified food, also makes it unlikely that the number of opportunities to disagree about appropriate product standards will decline in the future.

From the anecdotal evidence one can distil at least four stylised facts. First, in most cases the disputed policy regulates a product, the consumption of which causes local damages. Second, a key feature of such disputes is uncertainty about the environmental or safety benefits of the regulations. Rarely can the scientific consensus offer near certainty about the likely damages caused by a product, in particular in the early period after initial allegations of harm. Third, the general public tends to be less informed about the scientific evidence than the government agencies which are implementing policy. Basically there is an element of asymmetric information between the government and the public. Finally, even though the regulation is applied in a non-discriminatory way to both domestically-produced and imported goods the immediate cost impact of the regulation is usually nonuniform with typically higher costs falling on importers. In the beef hormones case the non-uniform impact is apparent. In other cases the new product standard also tends to change the relative cost structures of importers and domestic producers, usually in favour of domestic firms, which is the very basis of the trade barrier argument.

The main purpose of this paper is to develop a political economy model, which builds on these stylised facts and can explain how such disputes occur. A simple two good specific factors model is introduced in which one good causes local damages with some probability. Policy making is modelled in the spirit of the political agency literature, which views the political process as a principal-agent relationship in which voters have to provide incentives for their politicians. In line with the third stylised fact it is assumed that politicians have an informational advantage in evaluating the probability of damages from consumption of the affected good. It will be shown how a political equilibrium can emerge in which domestic politicians claim that the expected damages are higher than their foreign coun-

terparts do. This divergence can either be due to an excessively strict standard in the importing country or a too lax standard in the exporting country. Both cases constitute a political failure in the sense that the economy is off the second best Pareto frontier. A second purpose of the paper is to take a first step towards analysing the implications of institutional arrangements to settle or avoid trade disputes over product standards. The effects of two frequent proposals, mutual recognition of standards and harmonisation, are addressed in detail. My principal finding is that both proposals may not improve welfare relative to decentralised standard setting.

The possibility that product standards are (ab)used as "disguised barriers to trade" has been discussed widely among policy makers, but has so far received relatively little attention in the academic literature. Notable exceptions are Esty (1994), Laplante and Garbutt (1992), Sorsa (1995) and Engel (1999) which offer some definitions of "green protectionism" and informally discuss possible responses to it. There is also a growing literature on the political economy of environmental policy. Fredriksson (1997) and Aidt (1998) are two recent contributions, which analyse the choice of emission taxes in a small open economy within a common agency framework. Hillman and Ursprung (1992, 1994) introduce an environmental lobby into a lobby model of trade policy formation. The environmental lobby tries to influence trade policy as other environmental policy instruments are assumed to be absent. In contrast in the present paper distortions to the environmental policy instrument will be at the centre of the analysis.

The political agency approach to modelling the political process was pioneered by Barro (1973). It has been further developed by Ferejohn (1986), Austen-Smith and Banks (1989), Coate and Morris (1995) and Banks and Sundaram (1998). The model of Coate and Morris (1995) is closest to the present paper. Besley and Case (1995a, 1995b) provide empirical evidence from American state elections for the explanatory power of political agency models. Alesina, Baqir and Easterly (1998) and Alesina, Danninger and Rostagno (1999) present data on public employment which support the disguised transfer mechanisms modelled by Coate and Morris (1995).

The paper is also related to the recent literature that analyses the role of the

GATT/WTO in trade policy making. Two recent contributions to this literature are Bagwell and Staiger (1999) and Maggi (1999) which investigate the role that GATT/WTO has played in the post war reductions in tariff levels. This paper can be seen as an extension of this literature to the case of non-tariff barriers to trade.

The paper is organised as follows. Section 2 outlines the two country trade model in which policy-making takes place. Section 3 describes the political system. Section 4 analyses the equilibrium policy choices. Section 5 explores the implications of institutional interventions to settle or avoid disputes over product standards. Section 6 discusses some of the assumptions of the model and the final section concludes. All proofs are in the appendix.

2 The Model

The economy is modelled as a specific factors model. Two countries, home and foreign, trade a numeraire good n and good x. There are three factors of production, a specific capital stock for the domestic and foreign x industry and labour. Each country is populated by a politician, a representative consumer and a producer. All foreign variables will be denoted with an asterisk.

2.1 Preferences and Technology

The two countries are identical except for their endowment with specific capital and only the home country will be described. The domestic representative consumer supplies L units of labour and has the quasi-linear utility function

$$U = n + u(x) - D(\theta, x_F) \tag{1}$$

with the standard properties u' > 0 and u'' < 0. The damage function D will be described in detail in the next section. The consumer maximises n + u(x) subject to his budget constraint taking damages as given. The politician and producer for simplicity are assumed only to consume the numeraire good and not to suffer any damage D. The consumer price of good x is p and by the producer price is p_p .

The numeraire good is produced with a constant returns to scale technology which uses one unit of labour to produce one unit of output in both countries. Good x is produced with the technology

$$q_x = f(l_x, \overline{k}) \tag{2}$$

$$q_x^* = f(l_x^*, \overline{k}^*) \tag{3}$$

which is assumed to have constant returns to scale and diminishing returns to each factor. q_x and q_x^* denote output and l_x and l_x^* are the labour inputs into production of good x in the home and foreign country respectively. The specific capital stocks are denoted \overline{k} and \overline{k}^* and it is assumed that $\overline{k} > \overline{k}^*$. If additionally L and L^* are sufficiently large to ensure positive production in the numeraire sector in both countries, this implies that the foreign country has a comparative advantage in good x and will export it under free trade. The returns accruing to the two specific capital stocks are earned by the local producer in each country and will be denoted π and π^* .

2.2 The Externality

The key ingredient of the economic environment is the damages caused by the consumption of good x, which are modelled as a pure externality.² This externality has two important properties. First, it is only caused by the consumption of units of good x which come from foreign production, denoted x_F . This assumption is an extreme way of capturing the stylised fact introduced above that the impact of regulating the externality is generally non-uniform with higher costs falling on importers. In a more general model the externality would be caused by both

²Modelling the damage as an externality may seem puzzling for two reasons. First, as there is only one representative consumer, why doesn't she internalise the damage that she inflicts on herself? If I introduced many identical consumers that maximise $U = n + u(x) + D(\theta, mx_F)$ where m is the number of consumers, they could hardly influence damages and would therefore behave as my representative consumer. Second, in the beef hormones example damages seem to be born by the consumer directly, rather then through an externality. However, even in this case the health insurance system will spread the costs in form of a pecuniary externality. Furthermore similar results can be derived in a modified model where damages are not external but directly born by the consumer.

domestically and foreign-produced units of good x and the non-uniform impact would be generated by differing abatement costs between the domestic and foreign industry. The assumption here would then be the special case where abatement costs for the home industry are zero. Second, the damage of the externality is uncertain. Expected damage D in the home country is

$$D(\theta, x_F) = \theta x_F. \tag{4}$$

With probability θ consumption of x from foreign production causes constant marginal damages, which have by choice of units been normalised to one, and with probability $(1 - \theta)$ there is no damage. Damage is measured in units of the numeraire. Finally it is assumed that θ takes two discrete values θ_0 or θ_1 with $\theta_0 < \theta_1$. If the scientific consensus is θ_1 we are in the "high risk" state, whereas θ_0 will be called the "low risk" state. The assumption about the uncertainty is supposed to capture the idea that the scientific consensus only offers a probability estimate of the likely damage of the externality. As both states can result in the same damages, it is never possible to infer ex post the state of θ from the realized damages.

There also exists a simple abatement technology. The expected damage of foreign produced units of good x can be reduced to zero at constant costs of a units of labour per unit treated. The abatement activity can either be thought of as a separate sector or an extra production cost incurred by the foreign x industry for output shipped to a market that requires the application of the abatement technology.

2.3 Policy Instruments

There are two policy instruments in the first period. Politicians have to determine an environmental policy E and a lump sum transfer T. The environmental policy can take two forms. The first option N is to do nothing. The second option P is a product standard which requires the removal of the externality from all foreign-produced units of good x imported into the domestic market. To comply with this standard, the abatement technology needs to be operated. If the domestic politician, for example, chooses P while the foreign politician decides to do nothing,

then all foreign produced units of x that are exported to the home country are treated at per unit costs a while any units sold in the foreign market remain untreated. Finally the lump sum transfer $T \geq 0$ can be used to tax the consumer and transfer the proceeds to the local producer. In the second period the only available policy instrument is the transfer.

2.4 Welfare and Profits

The welfare consequences of implementing the product standard P in the home country can be evaluated with the following welfare function

$$W = CS + \pi - D(\theta, x_F) \tag{5}$$

where CS is consumer surplus and π are the returns to the domestic specific factor. The implications of the environmental policy for welfare and the return of the specific factors are summarised in the following lemma:

Lemma 1 There exists a range of values for θ_0 , θ_1 and a such that: i.) in state θ_1 it is welfare maximising to implement the product standard in both countries and not to do so in state θ_0 . ii.) Implementing P in state θ_1 increases the home consumers utility and implementing N^* in state θ_0 increases the foreign consumers utility. iii.) Implementing P increases π and implementing N^* increases π^* .

The intuition for the lemma is as follows. Implementing the product standard in the home country has effects very similar to a tariff where all tariff revenue is lost. All imports of x now have to be treated at per unit cost a which increases p and p_p in the home country. This increase in import prices is, on its own, welfare reducing. Even though the return to the domestic specific factor increases, this is outweighed by the reduction in consumer surplus as the home country imports good x. The benefit of the abatement activity is that expected damages, which are borne by the consumer, are reduced to zero.

For a given size of a there will therefore be a critical value $\tilde{\theta}_H$, below which expected damages are so small, that it is not optimal to operate the abatement technology. Similarly above this critical value expected damages are large enough

to make it worthwhile to operate the abatement technology. Furthermore increasing θ above $\tilde{\theta}_H$ will at some point result in expected damages that are so large that they outweigh the reduction in consumer surplus caused by the operation of the abatement technology. In this case implementing the product standard not only increases welfare, but also increases the domestic consumers utility.

The intuition for the foreign country runs along the same lines. Requiring the use of the abatement technology for sales in the foreign market imposes ceteris paribus a net cost on the foreign economy, which in this case takes the form of both a lower consumer surplus and a lower income of the foreign specific factor. For values of θ above a critical value $\tilde{\theta}_F$ expected damages are so high that the benefit of avoiding them outweighs the costs of operating the abatement technology. Below $\tilde{\theta}_F$ the opposite result holds. Furthermore reducing θ below $\tilde{\theta}_F$ will at some point make expected damages so small that implementing the abatement technology not only reduces welfare but also the consumers utility.

The critical values for the welfare effects of the standard, $\tilde{\theta}_H$ and $\tilde{\theta}_F$, will generally differ between the two countries due to standard terms of trade reasons. The product standard improves the home country's terms of trade as the import price of good x net of abatement costs declines. For the foreign country the product standard, however, results in a deterioration of the terms of trade. For the rest of this paper I assume that θ_0 , θ_1 and a are in the range characterised by lemma 1. In this range purely welfare maximising governments will implement the same policy in both countries. This result provides a very simple benchmark against which the outcome of the political equilibrium analysed in later sections can be compared. In the discussion I will return to the assumption of linear abatement and damage functions and show how relaxing them would change the analysis.

Finally a second more subtle implication of the assumption that parameters are such that lemma 1 holds is that the consumer always prefers the policy which also maximises social surplus. As the consumer will later be the decisive voter this coincidence of the consumers interest and welfare removes a straightforward tension from policy making.³

 $^{^3}$ Coate and Morris (1995) make an equivalent assumption to align the voter's preferences and social welfare.

3 The Political System

The political system is modelled in the tradition of the political agency literature. I consider a two period model. In the first period an incumbent implements policy in each country. At the end of the first period the two incumbents face an election in which the electorate in each country decides whether to re-elect the incumbent or elect a challenger. After the election second period policies are implemented and then the game ends. The structure of the domestic and foreign political system are identical. As before an asterisk denotes a foreign variable.

3.1 Politicians' Preferences

Similar to the contributions by Coate and Morris (1995) and Banks and Sundaram (1998) the model not only contains moral hazard, but also an element of adverse selection. Politicians come in two types, "good" and "bad". A good politician is for simplicity a very straightforward person. She always implements the welfare-maximizing policy and hence chooses P in the high risk state θ_1 and N in the low risk state θ_0 . Furthermore she never makes any cash transfers.⁴

A bad politician's preferences are more complicated. His per period utility function is $v_b(W, \pi + T)$ and he discounts second period utility at rate δ . The income of the local producer enters his utility not only as one component of welfare, but also as an additional argument. His utility is assumed to be strictly increasing in W and initially increasing but later decreasing in the second argument. In other words, a bad politician also cares about welfare, but would like to transfer at least some resources to the local producer. There are no explicit microfoundations for these preferences. One possible justification would be that the bad politician is susceptible to bribes from the well-organized producer lobby.

Lemma 1 has established how welfare and profits are affected by the environmental policy and the state θ . A bad domestic politician's preferences can

⁴The very simple modeling of the good politician could be relaxed at the cost of some more complexity as for example done in Coate and Morris (1995). However, all interesting equilibria of such a more general model involve the good politician maximising welfare and not doing any cash transfers, which would have to be insured through restrictions on the good politician's preferences and equilibrium refinements.

therefore be rewritten as $v_b(W(E, E^*, \theta), \pi(E, E^*) + T)$ or simply $v_b(E, E^*, T, \theta)$. There are three further assumptions about a bad politician's preferences. The first assumption concerns a bad incumbent's willingness to do harm.

Assumption 1 (bad politicians do harm) It is the case that $\psi_1 > 0$ and $\psi_1^* > 0$ with

$$\psi_1 = \{ v_b(P, N^*, 0, \theta_0) - v_b(N, N^*, 0, \theta_0) \}$$

$$\psi_1^* = \{ v_b^*(P, N^*, 0^*, \theta_1) - v_b^*(P, P^*, 0^*, \theta_1) \}.$$

This assumption says that given that no cash transfer has been made for a bad domestic (foreign) incumbent the increase in income of the local producer more than compensates the loss in welfare caused by the unwarranted (non) implementation of the environmental policy in the "low risk" state θ_0 ("high risk" state θ_1). The larger ψ_1 and ψ_1^* the stronger is a bad incumbent's incentive to implement the "wrong" environmental policy. If assumption 1 was violated, bad incumbents would not really be harmful, as they would never want to abuse the environmental policy.

The second assumption concerns a bad politician's propensity to make cash transfers. Let \widehat{T} be the cash transfer that maximizes a bad domestic politician's utility if domestic welfare is W and pre-transfer income of the domestic producer is π . That is:

$$\widehat{T}(W,\pi) = \arg\max v_b(W,\pi+T). \tag{6}$$

The resulting utility level will be $\hat{v}_b(E, E^*, \hat{T}, \theta)$. Similarly $\hat{T}^*(W^*, \pi^*)$ is the optimal transfer for a foreign bad politician.

Assumption 2 (the election matters) For $(E, E^*, \theta) \in \{P, N\} \times \{P^*, N^*\} \times \{\theta_0, \theta_1\}$ it is the case that $\psi_1 > 0$ and $\psi_1^* > 0$ with

$$\psi_2 = \min_{E, E^*, \theta} \left\{ \delta \widehat{v}_b(\widehat{T}) - \left[\widehat{v}_b(E, E^*, \widehat{T}, \theta) - v_b(E, E^*, 0, \theta) \right] \right\}$$

$$\psi_2^* = \min_{E, E^*, \theta} \left\{ \delta \hat{v}_b^*(\hat{T}^*) - \left[\hat{v}_b^*(E, E^*, \hat{T}^*, \theta) - v_b^*(E, E^*, 0^*, \theta) \right] \right\}.$$

The term $\hat{v}_b(E, E^*, \hat{T}, \theta) - v_b(E, E^*, 0, \theta)$ in square brackets is the utility that a bad domestic politician foregoes, if he does not implement the cash transfer in the first period. The assumption states that this has to be smaller than $\delta \hat{v}_b(\hat{T})$, which is the discounted value of his maximum utility in the second period, when he chooses the optimal transfer and there is no environmental policy. If this assumption was violated, bad politicians would not worry about the election. Their dominant strategy would be to always do the optimal cash transfer in the first period and be voted out of office with certainty. A sufficiently high pure rent from holding office would, for example, ensure that this assumption holds, so that bad incumbents do have an incentive to try to survive the election.

The final assumption is a more technical assumption which constrains a bad incumbent's rate of substitution between welfare and the income of his local producer at different levels of welfare and income of the local producer.

Assumption 3 ("single crossing") It is the case that

$$[v_b(P, P^*, 0, \theta_1) - v_b(N, P^*, 0, \theta_1)] \ge [v_b(P, N^*, 0, \theta_0) - v_b(N, N^*, 0, \theta_0)]$$
$$[v_b^*(N, N^*, 0^*, \theta_0) - v_b^*(N, P^*, 0^*, \theta_0)] \ge [v_b^*(P, N^*, 0^*, \theta_1) - v_b^*(P, P^*, 0^*, \theta_1)]$$

This assumption is similar to a single crossing assumption and states that for a bad domestic politician the increase in utility from implementing P rather than N is at least as large in the high risk state θ_1 than in the low risk state θ_0 even as the foreign incumbent's policy changes from N^* to P^* . As the foreign incumbent's environmental policy choice affects domestic welfare and profits through the terms of trade this inequality is not necessarily true for the general preferences of bad incumbents. Even though both profits and welfare increase in state θ_1 if P rather than N is chosen, these effects take place at a different point on the bad incumbent's utility function as the effects of switching from N to P in state θ_0 when $E^* = N^*$. If this assumption was not satisfied there would be the counter-intuitive possibility that a bad domestic incumbent derives more utility from implementing P in state θ_0 when N is optimal than from doing so in state θ_1 when P not only helps his producer, but also increases welfare.

3.2 Timing of Events and Information Structure

At the beginning of the first period nature moves and selects the type of politician in office in the first period. Their own type is only known to themselves and the other incumbent, but not to the voter in either country.⁵ Then voters in both countries receive a signal λ_I and λ_I^* with $0 < \lambda_I^*$, $\lambda_I < 1$ of the type of the incumbents which is imperfectly correlated with their type and which is also observed by both incumbents. The value of λ_I and λ_I^* is the probability that the domestic and foreign incumbent respectively are good and will be referred to as their initial reputations. Then nature moves again and selects the state θ . With probability ϕ , which is also common knowledge, the state is θ_1 and the externality is of "high risk" and with probability $(1-\phi)$ the state is θ_0 which is the "low risk" state. A crucial assumption is that the realization of θ is only observed by the two incumbents.

This informational asymmetry between the politicians and the electorate captures the idea that the incumbents have an advantage in assessing the scientific evidence on the riskiness of the externality. They have for example the possibility to commission a study which reveals the state θ but the results of this study cannot be credibly conveyed to the voter. Furthermore the assumption that both incumbents observe the state θ ensures that any disagreement between the politicians about the optimal policy is not caused by different information about the state of nature.

After having observed the state of nature the two incumbents simultaneously implement their environmental and transfer policies. At the end of the first period the realization of the externality is observed by voters in both countries and there is an election in both countries. The voters in this election are the representative consumer in the home and foreign country. The idea is that the producer is a small group which is not decisive in the election. First period policy choices and the realisation of the externality can be summarised in the track record (E, E^*, T, T^*, D) . Given the equilibrium strategies the voters will update their beliefs about the quality of their respective incumbent according to Bayes rule for any track record that

⁵If their type was private knowledge to each incumbent, then the results of Section 4 would still go through, but some of the extensions in Section 5 would need to be modified.

occurs with a positive probability. These updated beliefs will be denoted α and α^* . The election is a contest between the incumbent and a challenger. The domestic and foreign challenger have initial reputations λ_C and λ_C^* respectively, which are the voter's beliefs of the probability that the challengers are of the good type. The challengers' reputations are drawn from a common cumulative distribution function $G(\lambda)$ which is smooth and increasing and satisfies G(0) = 0. The winner of the election is in office in the second period and can choose to make a transfer. Then the game ends.

3.3 Definition of Political Equilibrium

A political equilibrium in this model is a perfect Bayesian equilibrium of the game between the two incumbents, the two representative voters and the two challengers. It consists of a strategy for each of the six players which is optimal given the other players' strategies. The strategies for each incumbent specify their policy choices in both periods as a function of the realization of θ and their own and the other incumbents type. The strategies for the representative voter in each country consist of two parts. The first part is a set of beliefs about the quality of the domestic incumbent generated by Bayes rule where possible. The second part is a rule that specifies when he will re-elect the incumbent. Finally a strategy for a challenger specifies his policy choices in the second period as a function of his type.

4 Analysis of Equilibrium

The policy game will be solved backwards in the usual way. The only available second period policy is the transfer. A bad domestic politician who is no longer constrained in his behaviour by an election will maximise his utility by implementing the optimal transfer \hat{T} and attain utility $\hat{v}_b(\hat{T})$. Similarly a bad foreign politician will chose \hat{T}^* and enjoy utility $\hat{v}_b^*(\hat{T}^*)$. A good politician in contrast will not do any transfers. Voters therefore clearly prefer to have a good politician in office in the second period.

4.1 The Voter's Optimal Re-election Rule

Given that the voter is better off with a good politician in office in the second period his optimal re-election rule is to re-elect the incumbent only if his updated beliefs about the incumbents quality are higher than the initial reputation of the challenger.⁶ As the initial reputation of the challenger in each country is drawn from the distribution $G(\lambda)$, the probability that the domestic incumbent will be re-elected is $G(\alpha)$ and $G(\alpha^*)$ for the foreign incumbent.

The updated beliefs α and α^* of the domestic and foreign voter about their respective incumbents are derived from the track record and the equilibrium strategies with the help of Bayes rule. If, for example, a bad domestic (foreign) politicians equilibrium strategy is to always choose $P(N^*)$ and not to make any transfer $T(T^*)$ and there turn out to be positive damages, denoted \overline{D} , then the beliefs of the domestic voter about the domestic incumbent will be:

$$\begin{array}{ll} \alpha(P,N^{*},0,0^{*},\overline{D}) & = & \\ \frac{\lambda_{I}\left[(1-\lambda_{I}^{*})\phi\theta_{1}\right]}{\lambda_{I}\left[(1-\lambda_{I}^{*})\phi\theta_{1}\right] + (1-\lambda_{I})\left[\lambda_{I}^{*}(1-\phi)\theta_{0} + (1-\lambda_{I}^{*})\left\{\phi\theta_{1} + (1-\phi)\theta_{0}\right\}\right]} \end{array}$$

The numerator of this expression is the probability that a good domestic incumbent would have generated this track record. The denominator is the probability that this track record is generated at all. Similar beliefs can be derived for the foreign incumbent and the other possible track records. As usual Bayes rule can only pin down beliefs on the equilibrium path. For equilibrium strategies of bad incumbents which do not involve transfers the observation of a transfer is an out of equilibrium event. Throughout the paper I will make the fairly mild assumption that out of equilibrium beliefs are such that a incumbent is believed to be bad with probability one if he makes a transfer.

The most interesting believes for this paper are the beliefs for the case where the domestic incumbent claims that the externality is of "high risk" and the foreign incumbent argues that it is of "low risk". If the underlying equilbrium strategy

⁶Even though this is how representative democracies are usually structured, this hides the deeper question why there are so few incentive instruments in the organisation of government. Tirole (1994) discusses this and related questions.

of bad domestic (foreign) politicians is to always choose $P(N^*)$ and not to make any transfers, then believes in this case have some intuitive properties which are summarized in the following lemma.

Lemma 2 Given strategies for a bad domestic (foreign) incumbent to always choose $P(N^*)$ and not to make any transfers and track records $(P, N^*, 0, 0^*, \overline{D})$ or $(P, N^*, 0, 0^*, 0)$ then voter's beliefs satisfy:

(i)
$$\partial \alpha/\partial \lambda_I > 0$$
, $\partial \alpha/\partial \lambda_I^* < 0$ and $\partial \alpha^*/\partial \lambda_I^* > 0$, $\partial \alpha^*/\partial \lambda_I < 0$
(ii) $\alpha(P, N^*, 0, 0^*, 0) < \alpha(P, N^*, 0, 0^*, \overline{D}) < \lambda_I$ and $\alpha^*(P, N^*, 0, 0^*, \overline{D})$
 $< \alpha^*(P, N^*, 0, 0^*, 0) < \lambda_I^*$.

The first part of the lemma says that in this case an incumbent's chances of re-election are increasing in his own initial reputation and decreasing in the other politician's initial reputation. The reason for this is that the two incumbents are exposed to yardstick competition. Voters observe both incumbents' policy choices and know that the same policy is appropriate in both countries. It is intuitive that disagreeing with the other incumbent becomes therefore more costly, in terms of a reduced re-election probability, if your voter's initial evaluation of the other incumbent's quality increases. The second part of the lemma states that such disagreement also imposes a reputational penalty on both incumbents relative to their initial reputations and that this penalty is bigger in the case where the realised damages do not correspond to the policy choice of the incumbent. The intuition for this is that disagreement can only arise if at least one incumbent is of the bad type. So the observation of disagreement makes it more probable that one's own incumbent is bad, but not certain as disagreements can also arise between a good incumbent in one country and a bad incumbent in the other country.

4.2 Payoffs to the Politicians

A bad domestic incumbent's per period utility is $v_b(W, \pi_1 + T)$. Normalising his utility of not being in office to zero his payoff over both periods V_b is:

$$V_b = v_b(E, E^*, T, \theta) + \begin{bmatrix} \theta G(\alpha(E, E^*, T, T^*, \overline{D}) \\ +(1 - \theta)G(\alpha(E, E^*, T, T^*, 0) \end{bmatrix} \delta \widehat{v}_b(\widehat{T})$$
 (7)

If the bad incumbent is in office in the second period he enjoys utility $\hat{v}_b(\hat{T})$ which is discounted by δ . Second period utility will only be realised if the incumbent is reelected, which happens with probability $G(\alpha(E, E^*, T, T^*, \overline{D}))$ if there are damages and with probability $G(\alpha(E, E^*, T, T^*, 0))$ if there are no damages. Finally first period utility as a function of the policy choices is $v_b(E, E^*, T, \theta)$.

4.3 Political Equilibrium in the First Period

The central result about first period policy-making is the following proposition:

Proposition 1 For sufficiently high values of $\psi_1, \psi_1^*, \psi_2, \psi_2^*$ there exists a range of initial reputations λ_I and λ_I^* in which a strategy for a bad domestic (foreign) incumbent always (never) to implement the product standard and both incumbents to not make any cash transfers is a PBE.

For the intuition of this proposition consider the incentives faced by a bad domestic incumbent in the "low risk" state θ_0 . For her implementing the product standard and doing the cash transfer are two alternative means of increasing the income of the domestic producer. Everything else equal she would prefer to use the transfer to increase the producer's income, as implementing the product standard is distortionary in the "low risk" state and even bad incumbents do care about welfare. The political system creates, however, incentives in the opposite direction. As implementing the transfer signals to the voter with certainty, that the incumbent is of the bad type, it entails certain electoral defeat. The electoral consequences of the product standard are less severe. Implementing it in the low risk state will cause a trade dispute as both good and bad foreign incumbents will not implement the product standard in the low risk state. Lemma 2 has established that a trade dispute imposes a reputational penalty on both incumbents in the form of a decreased re-election probability relative to their initial reputations, but re-election does not become impossible.

The political equilibrium characterised in proposition 1 differs markedly from the normative prescriptions derived in lemma 1. A bad domestic incumbent, for example, will implement the product standard even in state θ_0 , when doing so is not optimal on welfare grounds. I will define such an unwarranted implementation of the product standard "green protectionism". The environmental policy is implemented not because of its environmental benefits, but as an indirect way to transfer resources to the domestic x industry. Such "green protectionism" is a political failure in the sense of Besley and Coate (1998) as we are below the constrained second best utility possibility frontier. Instead of implementing the in state θ_0 inefficient environmental policy the bad politician could use the available transfer policy to pay the producer of good x the equivalent of the increase in profits that he enjoys under the product standard. This would leave the producer indifferent and make the consumer better off. The politician's first period utility would also increase as welfare increases while the producer's income stays the same. The available policy instruments could therefore achieve a Pareto improvement but the bad politician does not find it in his best interest to bring it about due to the different effects of the two policy instruments on his re-election probability.

Similarly a bad foreign incumbent will in the equilibrium of proposition 1 not implement the product standard even in the "high risk" state θ_1 . I will call this underprovision of environmental policy "environmental dumping" and it also constitutes a political failure in the sense defined above. Rather than exposing the foreign consumer to the "high risk" externality the foreign politician could implement the product standard and compensate the producer of good 2 for the fall in profits. This would increase the foreign consumer's utility and welfare and therefore also the foreign politician's first period utility.

When are we going to observe a trade dispute in which the foreign incumbent accuses the domestic politician of implementing an unnecessary product standard in the equilibrium of proposition 1? At least one of the incumbents has to be bad and has to engage in either "green protectionism" or "environmental dumping". Take the case of "green protectionism", which occurs in state θ_0 with a bad domestic incumbent in office. Both a good and a bad foreign incumbent will not implement the environmental policy in this state and hence there will be the described disagreement. Similarly in the case of "environmental dumping", which happens with a bad incumbent in office in the foreign country in state θ_1 , both

a good and bad domestic incumbent will implement the product standard in the domestic country in state θ_1 and the same dispute pattern will follow. To have a bad incumbent in office in the first period is on its own, however, not enough to generate a trade dispute. Take for example the combination of a bad domestic and a good foreign incumbent confronted with a "high risk" externality. In this case both incumbents will implement the product standard. The results of this discussion are summarised in the following corollary:

Corollary 1 A trade dispute involves at least one bad incumbent and either "green protectionism" or "environmental dumping".

In other words this corollary says that the two political failures, green protectionism and environmental dumping, are observationally equivalent from the perspective of the voter. If there is disagreement about the appropriate environmental policy, it is clear that one incumbent is abusing the environmental policy, but it is neither clear who is guilty nor whether environmental policy is over- or underprovided.

What is the importance of the yardstick competition between the two incumbents for the results derived so far? The yardstick competition is a constraint on bad incumbents' (mis-)behaviour. Disagreeing with a very reputable incumbent in the other country carries a high re-election penalty, as lemma 2 has established, and therefore becomes progressively less attractive. Suppose there was no yardstick competition, because voters could not observe policy choices in the other country. It can be shown that the equilibrium characterised in proposition 1 would then become an equilibrium in a larger parameter space. More precisely:

Corollary 2 In a world without yardstick competition there exists a larger range of initial reputations λ_I and λ_I^* in which the strategies of proposition 1 form an PBE equilibrium for any given values of $\psi_1, \psi_1^*, \psi_2, \psi_2^*$.

This corollary also allows me better to compare my proposition 1 to the results of Coate and Morris (1995). The model without yardstick competition is very close to their economic structure. My payoff structure differs slightly, but the same mechanism is operating. This corollary says that the inefficiencies that Coate and

Morris (1995) have identified are an equilibrium in a smaller range of parameters in the presence of yardstick competition between two incumbents, but they continue to exist.

5 Institutional Reform

With the model developed in the previous sections I will now address proposals to settle or avoid such trade disputes. Such interventions essentially add extra constraints to the decentralised policy making that has been analysed so far. In particular two possibilities will be discussed in some detail: Mutual recognition of standards and harmonisation. The question is whether these interventions can mitigate the political failure and improve welfare in the two countries.

5.1 Mutual Recognition of Standards

A common proposal to settle disputes over product standards is to agree on a mutual recognition of standards rule. The EU has, for example, gone a long way towards implementing this rule for intra-EU trade. A mutual recognition rule specifies that the product standard which applies to units of a good sold in the local market is also sufficient for exported units of the same good. In the context of this model this rule would specify that whatever product standard applies to units of good x sold in the foreign market will also be sufficient for exports to the home country. Some implications of this change in regime are summarised in the following proposition:

Proposition 2 A mutual recognition makes the implementation of the product standard less likely in the domestic country and has an ambiguous welfare impact relative to the equilibrium of proposition 1.

The intuition for this proposition is as follows. There are two opposing effects on the home country's welfare. Consider first the range of λ_I and λ_I^* in which

 $^{^7\}mathrm{See}$ Vogel (1995) for an extensive discussion of the EU's efforts to eliminate non-tariff barriers to trade and the role of mutual recognition of standards in this process.

proposition 1 applies. The positive effect of the mutual recognition rule is, that there will be no more "green protectionism". Whenever bad domestic incumbents would have implemented the product standard in state θ_0 there will now be no product standard as both good and bad foreign do not implement the standard in state θ_0 . The drawback of the mutual recognition rule is, however, that the domestic country is now going to suffer from the "environmental dumping" by the foreign country. In state θ_1 bad foreign incumbents will not implement the standard and there will therefore also be no standard in the home country. Without the mutual recognition rule both types of domestic incumbents would have implemented the standard in state θ_1 . The mutual recognition rule therefore essentially solves the problem of "green protectionism" but the domestic country now suffers from the "environmental dumping" in the foreign country. This intuition can be extended to initial reputations outside the range characterised in proposition 1. Which of these two effects dominates depends on the values of the abatement and damage costs and the probability with which bad politicians occur and it is not clear why one should dominate the other. The problem with mutual recognition is basically that it works on the implicit assumption that only politicians of the importing country can be bad, but that the political process of the exporting country is optimal, which is not true in this model. Mutual recognition is therefore not a promising intervention to increase welfare in both countries.

5.2 Harmonisation

Probably the most frequent approach to eliminating disputes over national product standards are attempts to harmonise standards. Both the European Commission and the WTO, for example, make major efforts in this direction. The traditional argument against harmonised standards is, that they do not take into account local differences in preferences and endowments. This is clearly a valid argument. However, to return to the example of the beef hormones again, it is hard to seen why citizens of two areas with similar per capita incomes like the United States and the EU should have a widely different willingness to pay to avoid damage from the consumption of growth hormones. So the inefficiency created by a uniform standard could be relatively small. In the economic environment of this paper the

costs of harmonisation are zero as the same standard is optimal in both countries.

It would therefore be very attractive to delegate the setting of product standards to an independent international body which is free from the pressures of the national political process and implements P in state θ_1 and N in state θ_0 . Unfortunately there is at the moment little evidence that any international body is sufficiently isolated from political pressure to achieve this outcome. In the absence of such an institution, efforts to harmonise standards will be the outcome of negotiations between national governments. A very simple way of thinking about such a negotiation is that voters in each of the two countries exert pressure on their respective incumbent to agree on a common policy with the other countries incumbent. One way to capture this idea in the model would be that voters agree to punish their incumbent with a reduction in his re-election probability, if he has failed to agree on a common policy. What would be the implications of such an incentive to harmonise? The next proposition summarises some possibilities:

Proposition 3 If voters punish failure to harmonise the product standard with a sufficiently strong decline in an incumbents re-election probability, then there exists both an equilibrium where the harmonised standard is on average too high and one in which the standard is on average too low.

The intuition for this proposition is as follows. I continue to assume that good incumbents will always implement the welfare maximising policy. If a bad incumbent is confronted with a good incumbent in the other country he therefore faces the choice of either implementing the welfare maximising policy or to see his re-election prospects decline. If both incumbents are bad there are two possible harmonisation levels. For both states of θ they could either both implement the product standard or both not do so. Suppose that a bad domestic incumbent's strategy is to implement N in state θ_0 and P in state θ_1 if the foreign incumbent is good and always implement N otherwise and never to make a transfer. Suppose

⁸One piece of anecdotal evidence for this claim is that scientific advisory panels, of for example the European Commission, are usually staffed with scientists from several countries, rather then a purely meritocratic selection. Furthermore observers tend to attach great importance to the question whether the scientists of a particular country had a dissenting opinion, which would be odd, if the deliberations were purely scientific.

that at the same time a bad foreign incumbent's strategy is to choose N in state θ_0 and P in state θ_1 if the foreign incumbent is good and N otherwise and also never to make a transfer. When will these strategies form an equilibrium? It is shown in the appendix that for sufficiently high λ_I and λ_I^* voters' updated beliefs about their incumbent's quality, if they observe harmonisation, are high enough to make a deviation to a first period transfer unprofitable for bad domestic and foreign incumbents. In this range of values for λ_I and λ_I^* leaving the harmonised "consensus" will also be unprofitable for bad incumbents regardless of the state θ , if the re-election penalty for non-harmonisation is sufficiently high as first period utility with a transfer is always at least as high as without a transfer.

In this equilibrium the product standard will always be at the optimal level unless two bad incumbents are in office in the first period and the state is θ_1 . In this case there will be no product standard, even though it would be welfare maximizing to have one. On average the standard will therefore be too low. The intuition for the case where the harmonised standard may be too high runs along the same lines. Even though harmonisation may be attractive in principle it is not clear that pressure to harmonise will result in efficient outcomes in the absence of institutions that help to implement the right level of harmonisation.

6 Discussion

This paper works with a very simple economic environment. In this section we will discuss one of the restrictive assumptions that have been made. The linear damage and abatement functions and restriction on parameters have allowed me to derive very simple welfare properties that I use as a benchmark for the evaluation of the political process. It has already been noted in the discussion of lemma 1 that the operation of the abatement technology affects the terms of trade. Even though the marginal costs of cleaning one unit of foreign-produced x is the same in both countries the effective cost of abatement differs between the two countries. In the case of the home country, for example, operation of the abatement technology reduces the price of imports and the net costs of using the abatement technology are therefore lower than the constant marginal cleaning costs a. In the foreign

country the terms of trade effect runs in the opposite direction. Implementing the abatement technology for the foreign market reduces foreign demand and hence lowers the world price of the foreign countries export good. Outside the range of parameters in which lemma 1 holds this could lead to a situation in which the home country finds it worthwhile to use the abatement technology in the "high risk" state while the foreign country does not do so.

More importantly the same terms of trade effects would also influence each country's optimal standard in a model with more general abatement and damage functions. Suppose for example that marginal damages remain constant but that damages from consuming foreign-produced units of good x can be reduced continuously by increasing the degree of abatement. In the case of the home country a tighter standard would not only reduce damages but also improve the home country's terms of trade. The home country's optimal degree of abatement would therefore reflect both these benefits. In contrast the foreign country's optimal standard would be biased downwards as tightening the standard has negative terms of trade implications. The political economy effects that I have identified would, however, continue to operate in such a more general model. The only difference would be that even two good incumbents would not choose the same policy in the two countries and the benchmark against which one would compare the political process would therefore be different.

There remains, however, an empirical question as to what the main reasons for the divergence of standards across countries are. There seems to be some anecdotal evidence that politicians not only choose different product standards to indirectly influence the terms trade, but rather because of some concrete political interests. However, in the absence of more empirical evidence this remains an open question.

7 Conclusion

This paper has addressed the growing number of trade disputes over product standards. A simple two country model has been developed in which a political failure in the national political process endogenously generates disagreements between the two politicians over the appropriate level of the product standard. In a second

step two common proposals to settle such trade disputes, mutual recognition of standards and harmonisation, have been analysed in the framework to determine, whether they can potentially mitigate the political failure. Both harmonisation and a mutual recognition rule were show to have ambiguous effects on the two countries' welfare relative to a system, where both countries independently implement their standards. Krugman (1997) has speculated that "the true purpose of international negotiations is arguably not to protect us from unfair foreign competition, but to protect us from ourselves." The two forms of international co-operation that have been analysed in this paper have, however, not found much evidence that this will be the case.

8 Appendix

Proof of Lemma 1:

There are four possible combinations of environmental policy across the two countries: (N, N^*) , (P, N^*) , (P, P^*) and (N, P^*) . Case (N, N^*) is free trade and the home country imports good x. In case (P, N^*) all imports to the domestic market are treated with the abatement technology. Assuming that the marginal abatement costs are not prohibitive, so that the foreign country continues to export good x arbitrage by consumers and producers will ensure that $p^* = p_p^* = p - a = p_p - a$. Substituting this into the world market clearing condition for good x gives:

$$d(p_p^* + a) - s(p_p^* + a) = s^*(p_p^*) - d^*(p_p^*)$$
(8)

where d and d^* are demand for and s and s^* are supply of good x in the domestic and foreign country respectively. Differentiating this identity yields:

$$\frac{\partial p_p^*}{\partial a} = -\frac{d' - s'}{d' - s' - s^{*'} + d^{*'}} < 0 \text{ and } > -1$$

Implementation of P therefore depresses p_p^* by less than one so that $p_p = p = p_p^* + a$ increase. It has already been argued in the main text that increasing the domestic prices is on its own welfare reducing for the domestic economy. These costs will, however, be worthwhile if the damages from imported units of x are large enough. Furthermore if expected damages are so large that they outweigh the loss in consumer surplus from implementing the product standard in the domestic country, then the policy is not only welfare increasing, but also improves the domestic consumers utility. Finally the return to the specific factor in the domestic x industry will unambiguously increase if p_p increases as wages remain equal to one due to the operation of the numeraire sector.

In case (P, P^*) foreign-produced units of good x have to go through the abatement technology regardless of their final destination. Arbitrage will now ensure that $p = p_p = p^* = p_p^* + a$. Substituting into world market clearing gives:

$$d(p) - s(p) = s^*(p) - d^*(p - a)$$

Differentiation yields:

$$\frac{\partial p}{\partial a} = -\frac{s^{*\prime}}{d' - s' - s^{*\prime} + d^{*\prime}} > 0 \text{ and } < 1$$

which implies that world prices of good x are higher than under free trade, but increase by less than a. It is also straightforward to check that p_p^* decreases by even more than in case (P, N^*) . As foreign wages are also always equal to one this implies that in case (P, P^*) the return of the foreign specific factor is lower than it is in the case (P, N^*) .

Furthermore consumer surplus declines as now $p^* = p_p^* + a$. The costs of implementing P^* are therefore a reduced consumer surplus and a reduced return of the specific factor. As already argued in the main text, these costs may or may not be outweighed by the reduction in expected damages from consuming untreated units of foreign-produced x. If expected damages are very small they will not even outweigh the loss in foreign consumer surplus and N^* will not only maximise welfare but will also increase the foreign consumer's utility.

Case (N, P^*) will not occur in equilibrium, but is nevertheless important as an outside option. In this case all foreign produced x that are sold on the foreign market first have to be treated with the abatement technology, while units that are exported to the home country remain untreated. A slightly simplified way of determining the effects of this policy combination is to assume that there is additionally a prohibition on imports of good x from the home country into the foreign market. In this case arbitrage will ensure that $p = p_p = p_p^* = p^* - a$. World market clearing now requires that:

$$d(p_p^*) - s(p_p^*) = s^*(p_p^*) - d^*(p_p^* + a)$$

Differentiation yields:

$$\frac{\partial p_p^*}{\partial a} = -\frac{d^{*'}}{d' - s' - s^{*'} + d^{*'}} < 0 \text{ and } > -1$$

Similar to case (P, P^*) producer prices in the foreign country fall, which implies a reduction in the return of the foreign specific factor and foreign consumer prices rise at the same time which reduces consumer surplus. The product standard therefore imposes also in this case costs on the economy, which may or may not be outweighed by the reduction in expected damages.

Relaxing the simplifying assumption that imports of x into the foreign country are prohibited does not change any of the results for case (N, P^*) , but the mechanism is different. If imports of clean domestic units of x are allowed into the foreign market, the foreign consumer would prefer to buy those rather than the more expensive foreign-produced units, that need to go through the abatement process first. The domestic consumer is at the same time indifferent between units of x from the domestic and foreign source. Suppose that the final outcome of this was that the entire output of the domestic x industry is shipped to the foreign country and the entire output of the foreign x industry goes to the home country. As demand for x is the same in both countries this would lead to a price difference between the countries equal in size to the autarky price difference. With a being non prohibitive, i.e. smaller than the autarky price, some output of the foreign x industry would therefore be treated with the abatement technology and sold in the foreign country, which implies that the same price structure prevails as before. This completes the proof.

Proof of Lemma 2:

Given the equilibrium strategies specified in the lemma the domestic and foreign voter's beliefs for the case where the domestic incumbent regulates the externality, but the foreign does not and neither makes any transfers are:

$$\alpha(P, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I (1 - \lambda_I^*) (\phi \theta_1)}{(1 - \lambda_I^*) \phi \theta_1 + (1 - \lambda_I) (1 - \phi) \theta_0}$$

$$\alpha(P, N^*, 0, 0^*, 0) = \frac{\lambda_I (1 - \lambda_I^*) \phi (1 - \theta_1)}{(1 - \lambda_I^*) \phi (1 - \theta_1) + (1 - \lambda_I) (1 - \phi) (1 - \theta_0)}$$

$$\alpha^*(P, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I^* (1 - \lambda_I) (1 - \phi) \theta_0}{(1 - \lambda_I) (1 - \phi) \theta_0 + (1 - \lambda_I^*) \phi \theta_1}$$

$$\alpha^*(P, N^*, 0, 0^*, 0) = \frac{\lambda_I^* (1 - \lambda_I) (1 - \phi) (1 - \theta_0)}{(1 - \lambda_I) (1 - \phi) (1 - \theta_0) + (1 - \lambda_I^*) \phi (1 - \theta_1)}$$

To demonstrate part (i) note that for $\alpha(P, N^*, 0, 0^*, \overline{D})$ and $\alpha(P, N^*, 0, 0^*, 0)$ the numerator is increasing in λ_I and the denominator in decreasing in λ_I . Similarly for $\alpha^*(P, N^*, 0, 0^*, \overline{D})$ and $\alpha^*(P, N^*, 0, 0^*, 0)$ the numerator is increasing in λ_I^* and the denominator is decreasing in λ_I^* . To establish the effect of λ_I^* on $\alpha(P, N^*, 0, 0^*, \overline{D})$ and $\alpha(P, N^*, 0, 0^*, 0)$ differentiation yields

$$\frac{\partial \alpha(P, N^*, 0, 0^*, \overline{D})}{\partial \lambda_I^*} = \frac{(-1)\lambda_I \theta_1 \phi(1 - \lambda_I)(1 - \phi)\theta_0}{[(1 - \lambda_I^*)\phi \theta_1 + (1 - \lambda_I)(1 - \phi)\theta_0]^2} < 0$$

$$\frac{\partial \alpha(P, N^*, 0, 0^*, 0)}{\partial \lambda_I^*} = \frac{(-1)\lambda_I \phi(1 - \theta_0)(1 - \lambda_I)(1 - \phi)(1 - \theta_1)}{[(1 - \lambda_I^*)\phi(1 - \theta_1) + (1 - \lambda_I)(1 - \phi)(1 - \theta_0)]^2} < 0$$

Similarly the effect of λ_I on $\alpha^*(P, N^*, 0, 0^*, \overline{D})$ $\alpha^*(P, N^*, 0, 0^*, 0)$ is given by:

$$\frac{\partial \alpha^*(P, N^*, 0, 0^*, \overline{D})}{\partial \lambda_I} = \frac{(-1)\lambda_I^* \theta_0 (1 - \lambda_I^*) (1 - \phi) \phi \theta_1}{[(1 - \lambda_I)(1 - \phi)\theta_0 + (1 - \lambda_I^*) \phi \theta_1]^2} < 0$$

$$\frac{\partial \alpha^*(P, N^*, 0, 0^*, 0)}{\partial \lambda_I} = \frac{(-1)\lambda_I^* \phi (1 - \theta_0) (1 - \lambda_I^*) (1 - \phi) (1 - \theta_1)}{[(1 - \lambda_I)(1 - \phi)(1 - \theta_0) + (1 - \lambda_I^*) \phi (1 - \theta_1)]^2} < 0$$

To show (ii) write down $\alpha^*(P, N^*, 0, 0^*, \overline{D}) < \alpha^*(P, N^*, 0, 0^*, 0)$. After some manipulations this simplifies to $\theta_1 > \theta_0$ which is true by assumption. Similarly the inequality $\alpha(P, N^*, 0, 0^*, \overline{D}) > \alpha(P, N^*, 0, 0^*, 0)$ also simplifies to $\theta_1 > \theta_0$.

Proof of Proposition 1:

Beliefs for the case where the domestic incumbent implements P and the foreign incumbent N^* have already been derived for the proof of lemma 2. Beliefs for other possible first period track records on the equilibrium path are:

$$\alpha(N, N^*, 0, 0^*, \overline{D}) = \alpha(N, N^*, 0, 0^*, 0) = 1$$

$$\alpha(P, P^*, 0, 0^*, \overline{D}) = \alpha(P, P^*, 0, 0^*, 0) = \lambda_I$$

$$\alpha^*(N, N^*, 0, 0^*, \overline{D}) = \alpha^*(N, N^*, 0, 0^*, 0) = \lambda_I^*$$

$$\alpha^*(P, P^*, 0, 0^*, \overline{D}) = \alpha^*(P, P^*, 0, 0^*, 0) = 1$$

Track records $(N, P^*, 0, 0^*, \overline{D})$ and $(N, P^*, 0, 0^*, 0)$ are out of equilibrium events and therefore not constrained by Bayes rule. I will assume that beliefs in this case are equal to one, which is the most unfavourable assumption possible. Finally as noted in the main text beliefs for track records that involve positive transfers are assumed to be zero.

Define re-election functions h (h^*) for the domestic (foreign) incumbent:

$$h(E, E^*, T, T^*, \theta) = \theta G(\alpha(E, E^*, T, T^*, \overline{D})) + (1 - \theta)G(\alpha(E, E^*, T, T^*, 0))$$
(9)

$$h^*(E, E^*, T, T^*, \theta) = \theta G(\alpha^*(E, E^*, T, T^*, \overline{D})) + (1 - \theta)G(\alpha^*(E, E^*, T, T^*, 0))$$
(10)

which are a function of the indicated variables and λ_I and λ_I^* .

To establish the existence of the equilibrium two deviations from the equilibrium strategy have to be ruled out for each incumbent. First, given the other incumbent's strategy and the initial reputations λ_I and λ_I^* domestic (foreign) incumbents must prefer $P(N^*)$ to $N(P^*)$ regardless of the state θ and the other incumbents type. The payoff to the bad domestic incumbent of implementing P in state θ_0 , when both good and bad foreign incumbents implement N^* is:

$$v_b(P, N^*, 0, \theta_0) + h(P, N^*, 0, 0^*, \theta_0) \delta \hat{v}_b(\hat{T})$$
(11)

which has to be greater than the payoff of doing N which is $v_b(N, N^*, 0, \theta_0) + \delta \hat{v}_b(\hat{T})$. This can be rearranged to yield:

$$h(P, N^*, 0, 0^*, \theta_0) > 1 - \frac{v_b(P, N^*, 0, \theta_0) - v_b(N, N^*, 0, \theta_0)}{\delta \hat{v}_b(\hat{T})}$$
(12)

As $h(P, E^*, 0, 0^*, \theta_0) \geq 0$ there will always be a critical value of ψ_1 (as defined in assumption 1) above which this inequality holds. The same critical value of ψ_1 will ensure that inequality (12) also holds when the state is θ_1 or $E^* = P^*$. If $E^* = N^*$ and the state is θ_1 rather than θ_0 , then the left hand side of (12) increases while the right hand side is unchanged. For $E^* = P^*$ and $\theta = \theta_1$ (which is the only other possible combination of states and strategies) we have $h(P, P^*, 0, 0^*, \theta_1) = \lambda_I > h(P, N^*, 0, 0^*, \theta_0)$ and assumption 3 implies that the right hand side of (12) has not increased.

The same method can be used to rule out that a foreign bad incumbent prefers P^* to N^* in state θ_1 when both good and bad domestic incumbents implement P. The payoff from choosing N^* is:

$$v_b^*(P, N^*, 0^*, \theta_1) + h^*(P, N^*, 0, 0^*, \theta_1) \delta \hat{v}_b^*(\hat{T}^*)$$
(13)

This has to be larger than the payoff from choosing P^* which is $v_b^*(P, P^*, 0^*, \theta_1) + \delta \hat{v}_b^*(\hat{T}^*)$. This yields the following inequality:

$$h^*(P, N^*, 0, 0^*, \theta_1) > 1 - \frac{v_b^*(P, N^*, 0^*, \theta_1) - v_b^*(P, P^*, 0^*, \theta_1)}{\delta \hat{v}_b^*(\hat{T}^*)}$$
(14)

As $h^*(P, N^*, 0, 0^*, \theta_1) \geq 0$ a sufficiently large ψ_1^* will ensure that this inequality holds. As in the case of the domestic incumbent the same ψ_1^* will ensure that (14) also holds when E = P and $\theta = \theta_0$ or E = N and $\theta = \theta_0$. In the first case the left hand side of (14) increases, while the right hand side is unchanged. In the second case the left hand side also increases and assumption 3 ensures that the right hand side does not increase.

The second possible deviation from the equilibrium strategy would be for bad incumbents to make a cash transfer in the first period. For a bad domestic incumbent in state θ_0 the payoff under the equilibrium strategy is (11), which has to be larger than $\hat{v}_b(N, N^*, \hat{T}, \theta_0)$, the payoff of selecting a first period cash transfer. This implies that

$$h(P, N^*, 0, 0^*, \theta_0) > \frac{\widehat{v}_b(N, N^*, \widehat{T}, \theta_0) - v_b(P, N^*, 0, \theta_0)}{\delta \widehat{v}_b(\widehat{T})}$$
(15)

has to hold. Assumption 1 implies that $v_b(P, N^*, 0, \theta_0) > v_b(N, N^*, 0, \theta_0)$ and a sufficiently large ψ_2 (as defined in assumption 2) will therefore ensure that this inequality is satisfied. Inequality (15) will also hold for the same ψ_2 if $E^* = N$ and $\theta = \theta_1$ or $E^* = P^*$ and $\theta = \theta_1$. In both these cases assumption 2 implies that the right hand side is no larger and the left hand side will be larger, as re-election becomes more probable.

Consider a bad foreign incumbents incentives to make a cash transfer. In state θ_1 his payoff under the equilibrium strategy is (13) and his payoff under the optimal cash transfer is $\hat{v}_b^*(P, E^*, \hat{T}^*, \theta_1)$ which implies the following inequality:

$$h^*(P, N^*, 0, 0^*, \theta_1) > \frac{\widehat{v}_b^*(P, P^*, \widehat{T}^*, \theta_1) - v_b^*(P, N^*, 0^*, \theta_1)}{\delta \widehat{v}_b^*(\widehat{T}^*)}$$
(16)

Assumption 1 implies that $v_b^*(P, N^*, 0^*, \theta_1) > v_b^*(P, N^*, 0^*, \theta_1)$ and a sufficiently large ψ_2^* will therefore ensure that the inequality holds. As before (16) will also hold for the same ψ_2^* if E = N and $\theta = \theta_0$ or E = P and $\theta = \theta_0$, as in both cases the right hand side does not increase, while the left hand side increases. This completes the proof.

Proof of Corollary 2

Let $\alpha(E,0,D)$ denote the domestic voter's up-dated beliefs about the quality of his incumbent, if the voter can only observe the domestic policy choices and there is no transfer. Similarly let $\alpha^*(E^*,0,D)$ be the equivalent beliefs for the foreign voter. It is straightforward to show that $\alpha(P,0,D) = \alpha(P,N^*,0,0^*,D)$ if $\lambda_I^* = 0$ and that $\alpha(P,0,D)$ is increasing in λ_I . Similarly it can be shown that $\alpha^*(N^*,0^*,D) = \alpha^*(P,N^*,0,0^*,D)$ if

 $\lambda_I = 0$ and that $\alpha^*(N^*, 0^*, D)$ is increasing in λ_I^* . This implies together with lemma 2 that $\alpha(P, 0, D) > \alpha(P, N^*, 0, 0^*, D)$ if $\lambda_I > 0$ and $\alpha^*(N^*, 0^*, D) > \alpha^*(P, N^*, 0, 0^*, D)$ if $\lambda_I > 0$. Now suppose that there is a trade dispute, so E = P and $E^* = N^*$, and that for the set of initial reputations λ_I and λ_I^* and $\psi_1, \psi_1^*, \psi_2, \psi_2^*$ at least one of the inequalities (12), (14), (15) or (16) is binding in the case where policy choices are observable in both countries. If foreign policy is now made unobservable, then the improvement in voters' assessment of the quality of their respective incumbents will increase the left hand side of all of these inequalities while leaving everything else unchanged. For given values of $\psi_1, \psi_1^*, \psi_2, \psi_2^*$ the initial reputations λ_I and λ_I^* could therefore be even lower without violating any of the above inequalities. This completes the proof.

Proof of Proposition 2

If the domestic politician always follows the foreign incumbents decisions, then his policy choices do not convey any information. The foreign voter's beliefs are now:

$$\alpha^*(N^*, 0^*, \overline{D}) = \frac{\lambda_I^*(1 - \phi)\theta_0)}{(1 - \phi)\theta_0 + (1 - \lambda_I^*)\phi\theta_1}$$

$$\alpha^*(N^*, 0^*, 0) = \frac{\lambda_I^*(1 - \phi)(1 - \theta_0)}{(1 - \phi)(1 - \theta_0) + (1 - \lambda_I^*)\phi(1 - \theta_1)}$$

$$\alpha^*(P^*, 0^*, \overline{D}) = \alpha^*(P^*, 0^*, 0) = 1$$

Note that the beliefs for track records $(N^*,0^*,\overline{D})$ and $(N^*,0^*,0)$ are increasing in λ_I^* and weakly higher then the foreign voter's beliefs if he observes $(P,N^*,0,0^*,\overline{D})$ and $(P,N^*,0,0^*,0)$ under decentralised decision making. The analysis for proposition 1 therefore implies that a strategy for a bad foreign incumbent to never implement the product standard is now an equilibrium in an even larger range of λ_I^* for any given value of ψ_1^* and ψ_2^* . Outside this range of initial reputations there exist only mixed strategy equilibria in which a bad foreign politician must be indifferent between the pure strategies that he randomises over. For this to be possible not implementing the product standard must carry a re-election penalty as bad foreign politicians otherwise always prefers to do so. To create such a penalty foreign bad politicians must in equilibrium be more likely than a good politician to implement the product standard. The same argument can be used to establish that bad domestic politicians, who ceteris paribus prefer implementation of the product standard to non-implementation, must in any equilibrium implement the product standard more frequently than a good politician.

Replacing domestic policy making through a mutual recognition rule with foreign policy making therefore strictly reduces the frequency of environmental policy. Good foreign politicians implement the same standard as good domestic politicians but no standard in state θ_0 where bad domestic incumbents would at least sometimes implement the standard. Furthermore bad foreign incumbents will implement the product

standard even less frequently than good foreign incumbents. This implies that the product standard becomes less frequent in the domestic country. The welfare impact of this change is ambiguous as has already been argued in the main text.

Proof of Proposition 3

To show that the equilibrium can involve on average to high standards take the following strategies. Good incumbents continue to always implement P and P^* in state θ_1 and N and N^* in state θ_0 . Bad domestic incumbents always implement the P in state θ_1 and N state θ_0 if the foreign incumbent is good and N in both states if the foreign incumbent is bad and do not make transfers in either case. Bad foreign incumbents implement P^* in state θ_1 and N^* in state θ_0 if the domestic incumbent is good and N^* in both states if the domestic incumbent is bad and also do not make transfers. Given these strategies possible track records are $(N, N^*, 0, 0^*, 0)$, $(N, N^*, 0, 0^*, \overline{D})$, $(P, P^*, 0, 0^*, \overline{D})$ and $(P, P^*, 0, 0^*, 0)$. Updated beliefs for the two incumbents in these cases are:

$$\alpha(P, P^*, 0, 0^*, \overline{D}) = \frac{\lambda_I \phi \theta_1}{\lambda_I \phi \theta_1 + (1 - \lambda_I) \lambda_I^* \phi \theta_1} \ge \lambda_I$$

$$\alpha(P, P^*, 0, 0^*, 0) = \frac{\lambda_I \phi (1 - \theta_1)}{\lambda_I \phi (1 - \theta_1) + (1 - \lambda_I) \lambda_I^* \phi (1 - \theta_1)} \ge \lambda_I$$

$$\alpha(N, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I (1 - \phi) \theta_0}{(1 - \phi) \theta_0 + (1 - \lambda_I) (1 - \lambda_I^*) \phi \theta_1}$$

$$\alpha(N, N^*, 0, 0^*, 0) = \frac{\lambda_I (1 - \phi) (1 - \theta_0)}{(1 - \phi) (1 - \theta_0) + (1 - \lambda_I) (1 - \lambda_I^*) \phi (1 - \theta_1)}$$

$$\alpha^*(N, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I^* (1 - \phi) \theta_0}{(1 - \phi) \theta_0 + (1 - \lambda_I^*) (1 - \lambda_I) \phi \theta_0}$$

$$\alpha^*(N, N^*, 0, 0^*, 0) = \frac{\lambda_I^* (1 - \phi) (1 - \theta_0)}{(1 - \phi) (1 - \theta_0) + (1 - \lambda_I^*) (1 - \lambda_I) \phi (1 - \theta_1)}$$

$$\alpha^*(P, P^*, 0, 0^*, \overline{D}) = \frac{\lambda_I^* \phi \theta_1}{\lambda_I^* \phi \theta_1 + (1 - \lambda_I^*) \lambda_I \phi \theta_1} \ge \lambda_I^*$$

$$\alpha^*(P, P^*, 0, 0^*, 0) = \frac{\lambda_I^* \phi (1 - \theta_1)}{\lambda_I^* \phi (1 - \theta_1) + (1 - \lambda_I^*) \lambda_I \phi (1 - \theta_1)} \ge \lambda_I^*$$

Note that the up-dated beliefs are increasing in each incumbents own initial reputation. Furthermore they are also either increasing in the other incumbents initial reputation or weakly larger than the incumbents own initial reputation. Two possible deviations have to be ruled out: implementing a positive transfer in the first period and changing the environmental policy away from harmonisation. Payoffs for bad domestic and foreign incumbents under the cash transfer are $\hat{v}_b(E, E^*, \hat{T}, D)$ and $\hat{v}_b^*(E, E^*, \hat{T}, D)$ respectively

in the first period and zero for the second period. Assumption 2 implies that if reelection is probable enough then the extra first period utility of a cash transfer does not outweigh the loss of second period utility due to a certain defeat in the election. Given the beliefs derived above there must therefore be a critical level of λ_I and λ_I^* above which $h(N, N^*, 0, 0^*)$, $h(P, P^*, 0, 0^*)$, $h^*(N, N^*, 0, 0^*)$, $h^*(P, P^*, 0, 0^*)$ are sufficiently high to make a first period cash transfer unattractive. In this range of λ_I and λ_I^* both incumbents will also not deviate from the harmonised standard if the re-election penalty is sufficiently large. Suppose that failure to harmonise is punished with a certain defeat in the election. As the first period utility gain from implementing a transfer is always at least as large as the gain from changing the environmental policy, this deviation can not be optimal. The implication of this is that the harmonised standard is optimal as long as at least one good incumbent is in office and too low when two bad incumbents are in office and the state is θ_1 .

To show that the harmonised standards can also be too high take the same strategies as above with one change. If two bad incumbents are in office their strategy is now that both implement the product standard regardless of the state. Beliefs in this case are:

$$\alpha(P, P^*, 0, 0^*, \overline{D}) = \frac{\lambda_I \phi \theta_1}{\lambda_I \phi \theta_1 + (1 - \lambda_I)(1 - \lambda_I^*)(1 - \phi)\theta_0}$$

$$\alpha(P, P^*, 0, 0^*, 0) = \frac{\lambda_I \phi (1 - \theta_1)}{\lambda_I \phi (1 - \theta_1) + (1 - \lambda_I)(1 - \lambda_I^*)(1 - \phi)(1 - \theta_0)}$$

$$\alpha(N, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I (1 - \phi)\theta_0}{\lambda_I (1 - \phi)\theta_0 + (1 - \lambda_I)\lambda_I^*(1 - \phi)\theta_0} \ge \lambda_I$$

$$\alpha(N, N^*, 0, 0^*, 0) = \frac{\lambda_I (1 - \phi)(1 - \theta_0)}{\lambda_I (1 - \phi)(1 - \theta_0) + (1 - \lambda_I)\lambda_I^*(1 - \phi)(1 - \theta_0)} \ge \lambda_I$$

$$\alpha^*(N, N^*, 0, 0^*, \overline{D}) = \frac{\lambda_I^*(1 - \phi)\theta_0}{\lambda_I (1 - \phi)\theta_0 + (1 - \lambda_I^*)\lambda_I(1 - \phi)\theta_0} \ge \lambda_I^*$$

$$\alpha^*(N, N^*, 0, 0^*, 0) = \frac{\lambda_I^*(1 - \phi)(1 - \theta_0)}{\lambda_I^*(1 - \phi)(1 - \theta_0) + (1 - \lambda_I^*)\lambda_I(1 - \phi)(1 - \theta_0)} \ge \lambda_I^*$$

$$\alpha^*(P, P^*, 0, 0^*, \overline{D}) = \frac{\lambda_I^* \phi \theta_1}{\phi \theta_1 + (1 - \lambda_I^*)(1 - \lambda_I)(1 - \phi)\theta_0}$$

$$\alpha^*(P, P^*, 0, 0^*, 0) = \frac{\lambda_I^* \phi (1 - \theta_1)}{\phi (1 - \theta_1) + (1 - \lambda_I^*)(1 - \lambda_I)(1 - \phi)(1 - \theta_0)}$$

Beliefs in this case are again increasing in the incumbents own initial reputation and either also increasing in the other incumbents reputation or weakly larger than their own initial reputation. The same argument as in the case of downward harmonisation can therefore be applied to show that above a set of critical values for λ and λ_I^* deviations

from the equilibrium strategies are unprofitable. This establishes that a situation where the standard is either at its optimal level or too high, which happens when two bad incumbents are in office in state θ_0 , is also an equilibrium. This completes the proof.

References

- [1] Aidt, Toke S. (1998) "Political Internalisation of Economic Externalities and Environmental Policy", *Journal of Public Economics*, 69, pp. 1-16.
- [2] Alesina, Alberto, Baqir, Reza, and Easterly, William (1998), "Redistributive Public Employment", NBER Working Paper Nr. 6746.
- [3] Alesina, Alberto, Danninger, Stephan and Rostagno, Massimo V. (1999) "Redistribution through Public Employment: The Case of Italy", NBER Working Paper Nr. 7387.
- [4] Austen-Smith, David, and Banks, Jeffrey S. (1989) "Electoral Accountability and Incumbency" in Peter C. Ordeshook (ed.), *Models of Strategic Choice in Politics*, Ann Harbour: University of Michigan Press.
- [5] Bagwell, Kyle and Staiger, Robert W. (1999) "An Economic Theory of GATT", American Economic Review, 89(1), pp. 215-48.
- [6] Banks, Jeffrey S., and Sundaram, Rangarajan K. (1998) "Optimal Retention in Agency Problems" *Journal of Economic Theory*, 82(2), pp. 293-323.
- [7] Barro, Robert J. (1973) "The Control of Politicians: An Economic Model" *Public Choice*, 14, pp. 19-42.
- [8] Besley, Timothy and Coate, Stephen (1998) "Sources of Inefficiency in a Representative Democracy: A Dynamic Analysis" *American Economic Review*, 88(1), pp. 139-56.
- [9] Besley, Timothy, and Case, Anne (1995a), "Does Electoral Accountability Affect Economic Policy Choices? Evidence from Gubernatorial Term Limits", *Quarterly Journal of Economics*, 60(3), pp. 769-98.
- [10] Besley, Timothy, and Case, Anne (1995b), "Incumbent Behaviour: Vote-Seeking, Tax-Setting, and Yardstick Competition", American Economic Review, 85(1), pp. 25-45.
- [11] Coate, Stephen and Morris, Stephen (1995) "On the Form of Transfers to Special Interests" Journal of Political Economy, 103(6), pp.1210-35.
- [12] Engel, Eduardo (1999) "Poisoned Grapes, Mad Cows and Protectionism", NBER Working Paper Nr. 6959.

- [13] Esty, Daniel (1994) "Greening the Gatt: Trade, Environment and the Future" Washington: Institute of International Economics.
- [14] Fredriksson, Per G. (1997) "The Political Economy of Pollution Taxes in a Small Open Economy", *Journal of Public Economics*, 33, pp. 44-58.
- [15] Hillman, Arye L. and Ursprung, Heinrich W. (1992) "The Influence of Environmental Concerns on the Political Determination of Trade Policy" in Anderson, Kym and Blackhurst, Richard (eds.) The Greening of World Trade Issues, Univerity of Michigan Press, Ann Arbor.
- [16] Hillman, Arye L. and Ursprung, Heinrich W. (1994) "Greens, Supergreens and International Trade Policy: Environmental Concerns and Trade Policy" in Carraro, C. (ed.) *Trade, Innovation, Environment*, Kluwer Academic Publishers, Dordrecht.
- [17] Ferejohn, John (1986) "Incumbent Performance and Electoral Control" *Public Choice*, 50, pp. 5-25.
- [18] Krugman, Paul (1997), "What Should Trade Negotiators Negotiate About?", Journal of Economic Literature, 35(1), pp. 113-20.
- [19] Laplante, Benoit, and Garbutt, Jonathan (1992) "Environmental Protectionism" *Land Economics*, 68(1), pp. 116-19.
- [20] Maggi, Giovanni (1999) "The Role of Multilateral Institutions in International Trade Cooperation", American Economic Review, 89(1), pp. 190-214.
- [21] Sorsa, Piritta (1995), "Environmental Protectionism, North-South Trade and the Uruguay Round", IMF Working Paper.
- [22] Tirole, Jean (1994), "The Internal Organisation of Government", Oxford Economic Papers, 46(1), pp. 1-29.
- [23] Vogel, David (1995) "Trading Up: Consumer and Environmental Regulation in a Global Economy", Harvard University Press, Cambridge.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

485	G. Duranton V. Monastiriotis	Mind the Gaps: the Evolution of Regional Inequalities in the UK 1982-1997
484	H. G. Overman Y. Ioannides	Zipfs Law for Cities: An Empirical Examination
483	H. G. Overman Y. Ioannides	Cross Sectional Evolution of the US City Size Distribution
482	Y. Ioannides H. G. Overman	Spatial Evolution of the US Urban System
481	H. G. Overman	Neighbourhood Effects in Small Neighbourhoods
480	S. Gomulka	Pension Problems and Reforms in the Czech Republic, Hungary, Poland and Romania
479	S. Nickell T. Jones G. Quintini	A Picture of the Job Insecurity Facing British Men
478	C. Dougherty	Numeracy, Literacy and Earnings: Evidence from the National Longitudinal Survey of Youth
477	P. Willman	The Viability of Trade Union Organisation: A Bargaining Unit Analysis
476	D. Marsden S. French K. Kubo	Why Does Performance Pay De-Motivate? Financial Incentives versus Performance Appraisal
475	S. Gomulka	Macroeconomic Policies and Achievements in Transition Economies, 1989-1999
474	S. Burgess H. Turon	Unemployment Dynamics, Duration and Equilbirum: Evidence from Britain
473	D. Robertson J. Symons	Factor Residuals in SUR Regressions: Estimating Panels Allowing for Cross Sectional Correlation
472	B. Bell S. Nickell G. Quintini	Wage Equations, Wage Curves and All That

471	M. Dabrowski S. Gomulka J. Rostowski	Whence Reform? A Critique of the Stiglitz Perspective
470	B. PetrongoloC. A. Pissarides	Looking Into the Black Box: A Survey of the Matching Function
469	W. H. Buiter	Monetary Misconceptions
468	A. S. Litwin	Trade Unions and Industrial Injury in Great Britain
467	P. B. Kenen	Currency Areas, Policy Domains and the Institutionalization of Fixed Exchange Rates
466	S. Gomulka J. Lane	A Simple Model of the Transformational Recession Under a Limited Mobility Constraint
465	F. Green S. McIntosh	Working on the Chain Gang? An Examination of Rising Effort Levels in Europe in the 1990s
464	J. P. Neary	R&D in Developing Countries: What Should Governments Do?
463	M. Güell	Employment Protection and Unemployment in an Efficiency Wage Model
462	W. H. Buiter	Optimal Currency Areas: Why Does the Exchange Rate Regime Matter?
461	M. Güell	Fixed-Term Contracts and Unemployment: An Efficiency Wage Analysis
460	P. Ramezzana	Per Capita Income, Demand for Variety, and International Trade: Linder Reconsidered
459	H. Lehmann J. Wadsworth	Tenures that Shook the World: Worker Turnover in Russia, Poland and Britain
458	R. Griffith S. Redding J. Van Reenen	Mapping the Two Faces of R&D: Productivity Growth in a Panel of OECD Industries
457	J. Swaffield	Gender, Motivation, Experience and Wages

To order a discussion paper, please contact the Publications Unit Tel 020 7955 7673 Fax 020 7955 7671 Email info@cep.lse.ac.uk Web site http://cep.lse.ac.uk